Author’s response to reviews

Title: Attitudes to kidney donation among primary care patients in rural Crete, Greece

Authors:

Emmanouil K Symvoulakis (symvouman@yahoo.com)
Ioannis D Komninos (yiankom@yahoo.co.uk)
Nikos Antonakis (antonakisnikos@yahoo.gr)
Myfanwy Morgan (myfanwy.morgan@kcl.ac.uk)
Athanasios Alegakis (thalegak@med.uoc.gr)
Emmanouil Tsafantakis (etsafantakis@yahoo.gr)
Marios Chatziarsenis (chatz_mar1@yahoo.com)
Anastas Philalithis (tassos@med.uoc.gr)
Roger Jones (roger.jones@kcl.ac.uk)

Version: 3 Date: 16 January 2009

Author’s response to reviews: see over
Date: 16 January 2009

BMC PUBLIC HEALTH

Manuscript No: 1108705289207643

Title: Attitudes to kidney donation among primary care patients in rural Crete, Greece

Author(s): Emmanouil K Symvoulakis, Ioannis D Komninos, Nikos Antonakis, Myfanwy Morgan, Athanasios Alegakis, Emmanouil Tsafantakis, Marios Chatziarsenis, Anastas Philalithis and Roger Jones

Dear editor

We would like to submit the revised manuscript, entitled “Attitudes to kidney donation among primary care patients in rural Crete, Greece”, which has been corrected in response to the referee 1 and editor’s comments. We have carefully revised and rephrased the statistical analysis section in order to make it easier for the readers and we have introduced the requested information on ethics approval in the method section. References have been finalised (i.e. Ref 8) and adjusted to the Journal style. The manuscript has been checked according to the format of the Journal.

We are looking forward to hearing from you.

Yours sincerely,

Emmanouil K. Symvoulakis